Author’s response to reviews

Title: Antibiotic Impregnated Articulating Cement Spacer Maintained For 7 years In Situ For Two Stage Primary Total Knee Arthroplasty: A Case Report

Authors:

Yong-Beom Park (whybe1122@gmail.com)
Chul-Won Ha (chulwon.ha@gmail.com)
Jae Jang (7umor@naver.com)
Manyoung Kim (mmanyeong@naver.com)

Version: 3 Date: 11 Apr 2019

Author’s response to reviews:

Editor Comments:

1. Consent for Publication

Please clearly state in the "Consent for Publication" section of your Declarations that "Written informed consent was obtained from the patient for the publication of this case report and any accompanying images".

-> Thank you for your comment. We revised sentence according to your comment from "Informed consent for the publication of this case report and any accompanying images was obtained from the patient included in the study." to "Written informed consent was obtained from the patient for the publication of this case report and any accompanying images".

2. Clean manuscript

Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please also ensure that all figures, tables, and supplementary/additional files are cited in the text.

-> We uploaded a clean manuscript. Thank you.